Author's response to reviews

Title: Exogenous glycosaminoglycans coat damaged bladder surfaces in experimentally damaged mouse bladder

Authors:

Kimberly D. Kyker (Kimberly-kyker@ouhsc.edu)
Jean Coffman (Jean-coffman@ouhsc.edu)
Robert E. Hurst (Robert-hurst@ouhsc.edu)

Version: 3 Date: 22 March 2005

Author's response to reviews:

The suggested formatting changes were made. The suggestion by Reviewer 1 was taken. The Discussion section also was re-ordered to improve the flow without changing the meaning.